Author’s response to reviews

Title: Arthroscopic guided biopsy and radiofrequency thermoablation of a benign neoplasm of the tibial spines area: a treatment option

Authors:

  Carmine Zoccali (carminezoccali@libero.it)
  Giuseppe Teori (teori@tiscali.it)
  Nicola Salducca (salduccafi@u.it)
  Berardino Di Paola (dipaoladino@yahoo.it)
  Ezio Adriani (eadriani@libero.it)

Version: 5  Date: 6 February 2012

Author’s response to reviews: see over
The manuscript needs some language corrections before being published